Author’s response to reviews

Title: The Interaction between HIV Testing Social Norms and Self-Efficacy on HIV testing among Chinese Men Who Have Sex with Men: Results from an Online Cross-Sectional Study

Authors:

Peizhen Zhao (08tjzpz@163.com)
Li Liu (liuli080805@163.com)
Ye Zhang (cetacean.413@hotmail.com)
Huanhuan Cheng (544657716@qq.com)
Bolin Cao (caobolin@seshglobal.org)
Chuncheng Liu (chuncheng@seshglobal.org)
Cheng Wang (wangcheng090705@163.com)
Bin Yang (yangbin101@hotmail.com)
Chongyi Wei (chongyiwei@hotmail.com)
Joseph Tucker (jdtucker@med.unc.edu)
Weiming Tang (weimingtangscience@gmail.com)

Version: 4 Date: 15 Oct 2018

Author’s response to reviews:

We have revised the manuscript based on the comments from the editor.